Author's response to reviews

Title: Feasibility and costs of water fluoridation in remote Australian Aboriginal communities

Authors:

Jonathon P Ehsani (jpehsani@hotmail.com)
Ross Bailie (Ross.Bailie@menzies.edu.au)

Version: 2 Date: 15 March 2007

Author's response to reviews: see over
Via email

Anita Makri MA, MPH
Assistant Editor
BMC-series Journals
editorial@biomedcentral.com

Dear Anita,

Thank you for inviting resubmission of our paper (MS: 1979928173123119) on the feasibility and costs of water fluoridation in remote Australian Aboriginal communities.

We attach a list of the comments made by the reviewer and our responses to them. We also enclose a revised manuscript with these changes marked using ‘Track Changes’. In addition, we attach a letter from the Minister of Health of the Northern Territory Government outlining the measures taken since the submission of the feasibility study.

Sincerely

Jonathon P. Ehsani and Ross Bailie